# Amoxicillin duration and dose for communityacquired pneumonia in children: the CAP-IT factorial non-inferiority RCT

Sam Barratt,<sup>1</sup> Julia A Bielicki,<sup>2</sup> David Dunn,<sup>1</sup> Saul N Faust,<sup>3</sup> Adam Finn,<sup>4</sup> Lynda Harper,<sup>1†</sup> Pauline Jackson,<sup>5</sup> Mark D Lyttle,<sup>5,6</sup> Colin VE Powell,<sup>7,8</sup> Louise Rogers,<sup>9</sup> Damian Roland,<sup>10,11</sup> Wolfgang Stöhr,<sup>1</sup> Kate Sturgeon,<sup>1</sup> Elia Vitale,<sup>2</sup> Mandy Wan,<sup>12</sup> Diana M Gibb<sup>1</sup> and Mike Sharland<sup>2\*</sup> on behalf of the CAP-IT Trial Team and the PERUKI and GAPRUKI Networks

<sup>1</sup>MRC Clinical Trials Unit, University College London, London, UK

<sup>2</sup>Paediatric Infectious Diseases Research Group, Institute for Infection and Immunity, St George's University of London, London, UK

<sup>3</sup>NIHR Southampton Clinical Research Facility and Biomedical Research Centre, University of Southampton, University Hospital Southampton NHS Foundation Trust, Southampton, UK

<sup>4</sup>Bristol Children's Vaccine Centre, School of Population Health Sciences/School of Cellular and Molecular Medicine, University of Bristol, Bristol, UK

<sup>5</sup>Emergency Department, Bristol Royal Hospital for Children, Bristol, UK

<sup>6</sup>Faculty of Health and Applied Sciences, University of the West of England, Bristol, UK

<sup>7</sup>Paediatric Emergency Medicine Department, Sidra Medicine, Doha, The State of Qatar

<sup>8</sup>School of Medicine, Cardiff University, Cardiff, UK

<sup>9</sup>Research and Development Nursing Team, Birmingham Women's and Children's NHS Foundation Trust, Birmingham, UK

<sup>10</sup>Paediatric Emergency Medicine Leicester Academic (PEMLA) Group, University Hospitals of Leicester NHS Trust, Leicester, UK

<sup>11</sup>SAPPHIRE Group, Health Sciences, Leicester University, Leicester, UK

<sup>12</sup>Evelina Pharmacy, Guy's and St Thomas' NHS Foundation Trust, London, UK

\*Corresponding author msharland@sgul.ac.uk †In memoriam

**Declared competing interests of authors:** David Dunn reports grants from the National Institute for Health Research during the conduct of the study (RP-PG-1212-20006). Saul N Faust reports personal fees or grants from AstraZeneca plc (Cambridge, UK)/Medimmune (Gaithersburg, MA, USA), Sanofi SA (Paris, France), Pfizer Inc. (New York, NY, USA), Seqirus UK Ltd (Maidenhead, UK), Sandoz (Holzkirchen, Germany), Merck KGAA (Darmstadt, Germany), GlaxoSmithKline plc (Brentford, UK) and Johnson & Johnson

(Brunswick, NJ, USA) outside the submitted work. In addition, Saul N Faust received grants for contract commercial clinical trials, which were paid to Saul N Faust's institution (with no personal payment of any kind). Last, Saul N Faust is a member of the Health Technology Assessment Commissioning Committee (2017–22). Adam Finn reports grants from GlaxoSmithKline plc, Pfizer Inc., Novavax (Gaithersburg, MA, USA), Sanofi Pasteur (Lyon, France), VBI Vaccines Inc. (Cambridge, MA, USA), Janssen Pharmaceuticals (Beerse, Belgium), Valneva SE (Saint-Herblain, France) and JITSUVAX outside the submitted work.

Published November 2021

DOI: 10.3310/hta25600

### Plain English summary

#### The CAP-IT RCT

Health Technology Assessment 2021; Vol. 25: No. 60

DOI: 10.3310/hta25600

NIHR Journals Library www.journalslibrary.nihr.ac.uk

## **Plain English summary**

Pneumonia (an acute lung infection) is a common diagnosis in young children worldwide. To cure this, some children are given antibiotics, but we do not currently know the best amount (dose) to give and the ideal number of days (duration) of treatment.

Taking antibiotics causes changes in bacteria, making them more resistant to treatment. This may be affected by the dose and duration, and is important because resistant bacteria are harder to treat and could spread to other people.

Amoxicillin is the most common antibiotic treatment for children with pneumonia. CAP-IT (Community-Acquired Pneumonia: a protocol for a randomIsed controlled Trial) tested if lower doses and shorter durations of amoxicillin are as good as higher doses and longer durations, and whether or not these affect the presence of resistant bacteria.

In total, 824 children in the UK and Ireland with pneumonia participated. They received either high- or low-dose amoxicillin for 3 or 7 days following discharge from hospital. To ensure that neither doctors nor parents were influenced by knowing which group a child was in, we included dummy drugs (placebo).

We measured how often children were given more antibiotics for respiratory infections in the 4 weeks after starting the trial medicine. To check for resistant bacteria, a nose swab was collected before starting treatment and again after 4 weeks.

One in every eight participating children was given additional antibiotics. We found no important difference in this proportion between 3 days and 7 days of amoxicillin treatment, or between lower or higher doses. Although children's coughs took slightly longer to go away when they received only 3 days of antibiotics, rash was reported slightly more often in children taking 7 days of antibiotics. There was no effect of dose of amoxicillin on any of the symptom measurements. No effect of duration of treatment or dose was observed for antibiotic resistance in bacteria living in the nose and throat.

### **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA** programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 13/88/11. The contractual start date was in March 2016. The draft report began editorial review in October 2020 and was accepted for publication in June 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

Copyright © 2021 Barratt *et al.* This work was produced by Barratt *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk